Author’s response to reviews

Title: Ultrasonographic median nerve cross-section areas measured by 8-point “inching test” for idiopathic carpal tunnel syndrome: a correlation of nerve conduction study severity and duration of clinical symptoms

Authors:

Shu-Fang Chen (fangoe@cgmh.org.tw)
Cheng-Hsien Lu (chlu99@ms44.url.com.tw)
Chi-Ren Huang (suika68@cgmh.org.tw)
Yao-Chung Chuang (ycchuang@cgmh.org.tw)
Nai-Wen Tsai (naiwen@adm.cgmh.org.tw)
Chiung-Chih Chang (neuro099@adm.cgmh.org.tw)
Wen-Neng Chang (cwenneng@ms19.hinet.net)

Version: 4 Date: 17 December 2011

Author’s response to reviews: see over
To editorial board

Thank you for your mentions. We have revised this manuscript accordingly.

Major revisions (we require the author to make these changes)

References: Please make sure references are cited in numerical order. We also notice that reference 32 has been cited yet there is no reference provided for number 32. Please either provide the necessary details for reference 32 or remove the citation.

We have corrected this error in the revised manuscript.

Tables: Please do not use ranges when listing tables. Please list all the figures, for example: Tables 5, 6, 7, 8 and 9.

Thanks for your mention. We have corrected the numbers of tables and figures in the revised manuscript also have made a corresponding correction in the text.

Tables: table 4 has been divided into parts 4.1 and 4.2. It is important that tables are numbered in whole numbers in ascending numerical order: 1, 2 and 3 etc. Please renumber the tables and any references to them in the text. Alternatively, please merge the tables into one larger table.

We have renumber the tables and also made a corresponding correction in the text.

(Table 4.1 →Table 4, Table 4.2 →Table 5, Table 5→Table 6)

Figures: Figure legend 2 contains graphics. These will either need to be included within the image as a key, or written in text (e.g. "black square").

We have written the graphic symbols in the legend section.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Highlighting/tracking: Please remove all highlighting and tracking from the manuscript.

We have removed the highlight markers of previous revision.

Tables: please note that we are unable to correctly display merged cells where the merged cell crosses rows: please re-layout your table without these merged elements.

We have re-layout the tables.

Figures: Please change the title "Legends" to "Figure legends"

We have changed the tile accordingly in the revised manuscript.
Figures: The image file should not include the title (e.g. Figure 1... etc.) or figure number. The figures are numbered automatically in the order in which they are uploaded.
We have corrected these errors of figures in the revised manuscript.

Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures
We have made related correction.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
Thank you for your mention.